Author's response to reviews

Title: Ethnic minorities and prescription medication; concordance between self reports and medical records

Authors:

Ellen E Uiters (ellenbart@minjus.nl)
Liset L van Dijk (l.vandijk@nivel.nl)
Walter D Deville (w.deville@nivel.nl)
Marleen M Foets (m.foets@erasmusmc.nl)
Peter P Spreeuwenberg (p.spreeuwenberg@nivel.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)

Version: Date: 30 August 2006

Author's response to reviews:

Accompanying letter: Revisions manuscript "Ethnic minorities and prescription medication; concordance between self reports and medical records"

Authors: Ellen Uiters, Liset van Dijk, Walter Deville, Marleen Foets, Peter Spreeuwenberg, Peter Groenewegen

Dear editor,
Thank you for accepting our manuscript for publication. The following changes were made to our final manuscript.
Methods: the following was add to the text (page 4, line 25): The study was carried out according to Dutch legislation on privacy. The privacy regulation of the study was approved by the Dutch Data Protection Authority. According to Dutch legislation, obtaining informed consent is not obligatory for observational studies. Furthermore the information regarding ethical approval was removed from page 11.
Competing interests: the following was added to the text (page 11, line 8):: The authors declare that they have no competing interests.
Authors’ contributions: the following was added to the text (page 11, line 12) : All authors read and approved the final manuscript. Furthermore LvD was added as contributor to the statistical analyses.

The funding information was added to the acknowledgements section.

References were placed the references after the Acknowledgements.

Box 1 was changed into table 1. Subsequently the references were updated.

With kind regards,

Ellen Uiters